Author’s response to reviews

Title: Factors Associated with Prelacteal Feeding Practices in Debre Berhan District, North Shoa, Central Ethiopia: A Cross Sectional Community Based Study

Authors:

Mesele Argaw (mdamte5@gmail.com)
Maeza Asfaw (maezymitiku@gmail.com)
Mekonen Ayalew (mokiebir@yahoo.com)
Binyam Desta (binyam_desta@jsi.com)
Thandisizwe Mavundla (mavuntr@unisa.ac.za)
Kassa Gidebo (kassadaka@yahoo.com)
Aynalem Frew (aynhm@yahoo.com)
Aychiluhim Mitiku (aychidamte@gmail.com)
Alebel Dessale (alebel.yaregal@gmail.com)

Version: 3 Date: 24 Jan 2019

Author’s response to reviews:

Author’s response to reviewer

Factors Associated with Prelacteal Feeding Practices in Debre Berhan District, North Shoa, Central Ethiopia: A Cross-Sectional Community-Based Study

Authors:

Mesele Damte Argaw1,2,*, Maeza Mitiku Asfaw3, Mekonen Birhane Ayalew 4, Binyam Fekadu Desta 1,2 , Thandisizwe Redford Mavundla2, Kassa Daka Gidebo5 , Aynalem Hailemichael Frew1, Aychiluhim Damtew Mitiku1 , Alebel Yaregal Desale6

Version: 3 Date: 24 January 2019
Dear Editors of BMC Nutrition Journal

Thank you so much for considering our paper in title “Factors Associated with Prelacteal Feeding Practices in Debre Berhan District, North Shoa, Central Ethiopia: A Cross-Sectional Community Based Study” (NUTN-D-18-00202R2) for publication on the Journal of BMC Nutrition. We want to thank all editorial team members and reviewers for their invaluable comments and suggestions on our manuscript. All comments have been addressed properly in the revised paper detail response to all comments follows below.

On behalf of all authors,

Yours Sincerely,

Mesele Damte Argaw, MPH
Senior Health Systems Strengthening Advisor
USAID Transform: Primary Health Care, Ethiopia
JSI Research & Training Institute Inc.
P.O. Box 1392 code 1110, Addis Ababa, Ethiopia
Comment 1. In your List of abbreviations, please provide the unabbreviated definition of HSTP.

Comment Accepted: on page 11 line 337, the expended form of HSTP was included in the list of Acronyms.

Comment 2. In your Ethics approval and consent to participate subsection of the Declarations, please clarify whether participants provided written or verbal consent to participate in your study.

Comment accepted: on page 11 line 348, the sentence was rephrased as presented below.

Comment 3. Submission of a manuscript to a BioMed Central journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality. Thus, we request that you upload English language versions of all questionnaires developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them.

Comment Accepted: The English version of questionnaire was attached as supplementary files.

Comment 4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Comment Accepted: a clean file as per the above instruction prepared and attached

Thank you so much

On behalf of all the authors

Mesele Damte Argaw